Author’s response to reviews

Title: Association of body mass index with amnestic and non-amnestic mild cognitive impairment risk in elderly

Authors:
Feng Wang (wangfeng0921@126.com)
Minghui Zhao (minghui5468@163.com)
Zhaoli Han (zhaoli8525@163.com)
Dai Li (dai528087@163.com)
Shishuang Zhang (zyyxhk@126.com)
Yongqiang Zhang (emaplecde@126.com)
Xiaodong Kong (Miao7698@126.com)
Ning Sun (sun75669@126.com)
Qiang Zhang (zhangqiang_yulv@163.com)
Ping Lei (leiping_1974@163.com)

Version: 6  Date: 04 Sep 2017

Author’s response to reviews:

Dear editor

Thank you for your great effort! All questions have been carefully answered. We hope the revised article will be acceptable! Thank you very much!

Sincerely yours,

Wang Feng, Lei Ping

-----------------------------------------------
1. In your Ethics approval section, please clarify whether informed consent obtained from the guardians of the participants had been written or verbal.

For those with cognitive impairment, we got the written informed consent from their legal guardians. (Line 310)

2. Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript.

Yes, "Not applicable" has been added. (Line 315)

3. Thank you for amending the Funding statement, please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The funding body had no involvement in any stage of the study. (Line 323)

4. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

Yes, the article has been revised by “American Journal Experts”. We tried our best to improve the language. We hope the revised article will be acceptable! Thank you very much!